Author's response to reviews

Title: Parkinsonism secondary to metastatic lesions within the central nervous system: a case report

Authors:

Eduardo Hortelano (edhortel@hotmail.com)
Christian Perea (cperea@gmail.com)
Esther Uña (aunacid@hotmail.com)
Amelia Cebayos (ameliac@gmail.com)
Patricia Diezhandino (patridiezhandino@gmail.com)
Montserrat Gonzalez (montgonza@yahoo.es)

Version: 5 Date: 4 March 2010

Author's response to reviews: see over
4\textsuperscript{th} of February, 2010

Dear sir,

I have corrected the manuscript entitled “Parkinsonism secondary to metastatic lesions within the central nervous system: a case report” as you have suggested.

Esther Uña Cidón, MD, PhD and Professor

Medical Oncology Department

Clinical University Hospital

Valladolid (Spain)